Author's response to reviews

Title: Stroke prevalence in Spanish elderly from screening surveys

Authors:

Raquel Boix (rboix@isciii.es)
Jose Luis del Barrio (jlbarrio@isciii.es)
Pedro Saz (psaz@unizar.es)
Ramon Rene (rrene@csub.scs.es)
Jose Maria Manubens deecesed (FMANUBENS@telefonica.net)
Antonio Lobo (loboa@hcu-iblesa.es)
Jordi Gascon (jordigneuro@csub.scs.es)
Ana de Arce (med002273@saludalia.com)
Jaime Diaz-Guzman (jdiazg@meditex.es)
Alberto Bergareche (abergarechey@meditex.es)
Felix Bermejo-Pareja (fbermejop2004@yahoo.es)
Jesus de Pedro-Cuesta (jpedro@isciii.es)

Version: 8 Date: 28 September 2006

Author's response to reviews: see over
Dear Editorial Production Team:

We have done all formatting changes requested:

Short title and institute sections from the title page have been removed.

The authors' qualifications (e.g. PhD / MD) have been removed from the submissions system.

The Spanish Epidemiological Study Group on Ageing to be a separate group of authors, has been included this as the final author in the submission system (then acknowledge the authors (of the group) in the Acknowledgments section).

Sentence case has been used for all headings and sub-headings in the manuscript.

The keywords section has been removed.

Introduction has been renamed 'Background'.

Conclusions section has been insert after the Discussion.

References - The reference list contain all authors' names

Table 1 – has been submitted in portrait format.

Website addresses - Websites have been removed from the text, and cited in the main reference list only.

Yours sincerely,

Dra. R. Boix